Author's response to reviews

Title: Aortic distensibility measured by pulse-wave velocity is not modified in patients with Chagas Disease

Authors:

Humberto Villacorta (huvillacorta@globo.com)
Luiz A Bortolotto (hipluiz@incor.usp.br)
Edmundo A Fernandez (earteaga@incor.usp.br)
Charles Mady (charles.mady@incor.usp.br)

Version: 4 Date: 12 June 2006

Author’s response to reviews:

The following formatting changes were provided:
1. References were cited in the text using numbers in square brackets.
2. The reference list was provided containing all authors’ names.
3. All page numbers were written in full: e.g. 223-227 NOT 223-7.

Best Regards

Humberto Villacorta